## **Economics and Finance in Indonesia**

Volume 65 Number 1 *June 2019* 

Article 5

8-2-2019

# Microcredit Accessibility in Rural Households: Evidence from Indonesia

Danang Budi Santoso

Faculty of Agribusiness and Commerce, Lincoln University, New Zealand. Ministry of Agriculture, Republic of Indonesia, danangkln@gmail.com

Follow this and additional works at: https://scholarhub.ui.ac.id/efi

Part of the Finance Commons, Macroeconomics Commons, Public Economics Commons, and the Regional Economics Commons

#### **Recommended Citation**

Santoso, Danang Budi (2019) "Microcredit Accessibility in Rural Households: Evidence from Indonesia," *Economics and Finance in Indonesia*: Vol. 65: No. 1, Article 5. DOI: 10.47291/efi.v65i1.635 Available at: https://scholarhub.ui.ac.id/efi/vol65/iss1/5

This Article is brought to you for free and open access by UI Scholars Hub. It has been accepted for inclusion in Economics and Finance in Indonesia by an authorized editor of UI Scholars Hub.

Economics and Finance in Indonesia Vol. 65 No. 1, June 2019 : 67–88 p-ISSN 0126-155X; e-ISSN 2442-9260

## Microcredit Accessibility in Rural Households: Evidence from Indonesia

Danang Budi Santoso<sup>a,b,\*</sup>, and Christopher Gan<sup>a</sup>

<sup>a</sup>Faculty of Agribusiness and Commerce, Lincoln University, New Zealand <sup>b</sup>Ministry of Agriculture, Republic of Indonesia

#### Abstract

Extant literature suggests that microcredit may improve borrowers' social and economic welfare. This study aimed to investigate the microcredit accessibility of rural households in Indonesia. Binary Logistic regression was used to distinguish the characteristics of microcredit borrowers and non-borrowers, and the factors affecting the approval of microcredit application. Primary data was collected through a survey on 488 rural households in Bantul, Yogyakarta. The empirical results suggested that age, marital status and education attainment significantly affect the characteristics of clients and non-clients of microcredit. The results also revealed that age of borrowers, household income, interest rates, and loan duration are key determinants affecting microcredit accessibility. The implications of the findings were provided. **Keywords:** microfinance; microcredit access; Indonesia; logit model

JEL classifications: G21; G18; H81

## 1. Introduction

#### 1.1. Background

The world has been encouraged by the socioeconomic impacts of microfinance on poverty alleviation, most notably in less-developed countries. A microfinance institution (MFI) is described by Getubig and Gibbons (2000) as the provision of intermediation by a financial organization through the distribution of small loans, the acceptance of small savings and the provision of other financial products and services to the poor. The main objective of microfinance is to effectively and deliberately reduce or eliminate poverty within in a reasonable time by providing the poor with access to microcredit (Quinones & Remenyi 2000). In this regard, poor households are treated as potential borrowers in

\* *Corresponding Author*: Ministry of Agriculture, Republic of Indonesia, Jl. RM Harsono No. 3, Ragunan, Jakarta 12550, Indonesia. Email: danangkln@gmail.com.

such a way that they are able to set up their own small businesses and could then escape from the poverty trap.

67

In contrast, conventional banks have some of the most stringent requirements when they deal with the underserved/poor groups who lack collateral and exhibit poor credit-worthiness, those sometimes known as "un-bankable." Thus, formal financial institutions (FIs), such as commercial banks and rural banks, which traditionally serve banks' clients, are reluctant to serve the poor mainly because they fail to meet the selection criteria, such as the physical collateral set by the financial institutions (Li, Gan, & Hu 2011a).

Amidst worrying reports about poor households who live under the poverty line<sup>1</sup>, microfinance programs for poverty alleviation has been burgeoning in developing countries, such as in Bolivia,

<sup>&</sup>lt;sup>1</sup>Based on the World Bank definition, the poverty line is defined as people who are living on less than US\$1.90 a day at 2015 international prices (purchasing power parity / PPP). Source: http://data.worldbank.org.

Bangladesh and Indonesia. The microfinance development by building a set of inclusive financial institutions, has raised the hope that much poverty can be alleviated. As a result, economic and social structures can be transformed at the grass root level by providing financial services to low-income households (Morduch 1999).

As Hermes and Lensink (2011) have argued, access to finance may contribute to long-lasting increases in income from a rise of investment in income-generating activities and a possible diversification of sources of income for low-income groups, particularly rural households.

Microfinance enables rural households to accumulate assets, smooths consumption in times of economic shocks, reduces their vulnerability due to illness, drought and crop failures, and achieves better education, health and housing outcomes for the borrowers' households. In addition, access to finance may contribute to an improvement in the social and economic position of women in family decision-making. Microfinance may also have positive spill-over effects as its impact surpasses just the economic and social improvement of the borrowers. For example, microcredit borrowers are likely to obtain higher incomes per capita and other social security protection, such as better education, health and housing. Furthermore, the positive assessment that microfinance contributions help to reduce poverty has convinced many governments, NGOs, and individuals to support the development MFIs and their activities (see for example, Hermes & Lensink 2011, Saad & Duasa 2011, Morduch 1998).

The success story of microfinance cannot ignore the impact from the Grameen Bank in Bangladesh since its establishment in 1970's. The bank was founded by Dr. Muhammad Yunus because of his passion for helping poor people, especially women, by providing small and soft loans from his own pocket to enable those villagers to buy materials for projects such as bamboo weaving and producing pots. In recognition of his innovation, he was awarded the Nobel Peace Prize in 2006 (Johnston & Morduch 2008). The Grameen Bank practice is to use a "group lending" and "joint liability" scheme. It was Yunus's idea that before borrowers receive loans, they should voluntarily form a group. Albeit the microcredit loans are made individually, all members of the group would be responsible for the loan repayments. The groups consist of five borrowers; the first two receive loans, then to the next two, and then the fifth borrower. These groups of five members meet weekly with seven other groups, so bank staff will thus meet with forty clients at a time. According to the rules, if one member defaults, all members in the group are denied subsequent loans (Morduch 1999).

### 1.2. Research Problem

Indonesia has a population of about 250 million people and is considered a lower middle-income country, with 17.4% of the population regarded as poor (31.2 million)<sup>2</sup>. About 214 million people (20%) still depend on micro and small-scale businesses for their living, but only 10 million of the 42 million microenterprises have access to credit from formal financial institutions (Banking With The Poor 2013). A survey conducted in 2002 found that 40% of poor households were judged to be creditworthy based on the criteria of loan officers, but fewer than 10% of poor households had borrowed from a formal micro-bank (Johnston & Morduch 2008).

The 1997 Asian financial crisis also affected Indonesian financial sectors. The central banks of affected Asian countries were not immune from the crisis, and tried to mitigate the risk of future

<sup>&</sup>lt;sup>2</sup>Source: World Bank Data - http://data.worldbank.org/ country/indonesia.

Economics and Finance in Indonesia Vol. 65 No. 1, June 2019

bank failures by promulgating a series of regulatory reforms. Inevitably, these reforms also changed Indonesian's microfinance outlook, which became larger and centralized by re-regulations to amalgamate relatively small, community-based financial institutions which were considered suspicious and hostile (Rosengard et al. 2007). According to Rosengard et al. (2007), financial reforms in Indonesia have concentrated on the default banks rather than mitigating banking risks and decreasing the access of low-income households and enterprises to formal financial services, especially in rural areas. Furthermore, the Indonesian government has also weakened incentives for innovation and outreach at the micro-banking level. The policy makers have concentrated credit risk by standardizing the banking system such as establishing mandatory village MFIs and converting government MFIs to the People Credit's Bank (BPR), weakening or severance of provincial government oversight and technical support by the provincial government, the centralization of operations, and a preference for standard loan products and delivery systems (Rosengard & Prasetyantoko 2011).

Rosengard and Prasetyantoko (2011) have argued that the financial sectors in Indonesia are currently characterized by two perplexing paradoxes:

- Indonesian microfinance institutions have been successful in their outreach and innovation for the past 25 years, but accessibility to microfinance services is an on-going problem for many poor households.
- 2. Indonesia's commercial banks are regarded as liquid, solvent, and profitable, and the Indonesian economy has been doing reasonably well over the past decade, but small and medium enterprises (SMEs) face a credit crunch.

Based on the current problems of credit accessibility to Indonesian rural households, this study aims to investigate the credit accessibility on rural households in Indonesia. Moreover, Indonesia is still improving to become middle incomes country. However, heavily prevalence of poverty in the rural areas and lacking access of formal and informal credit to rural households has been considered as hurdles in improving the livelihood for rural households.

To address the aforementioned research problems, the research questions are as follows:

- What are the characteristics of rural household members who use microcredit loans versus those who do not use microcredit loans in Indonesia?
- 2. What are the determinants of credit accessibility to rural households in Indonesia?
- 3. What are the best practices of microfinance programs for the policy decision makers?

There will be two contributions of this paper. First, the study will identify the determinants affecting microcredit accessibility of rural households in Indonesia, bridging the gap in the Indonesian microfinance literature by using an empirical approach comprising a field survey and a structured questionnaire. Second, this study will propose some policy recommendations for Indonesian policy makers that may yield better strategies to help expand their microcredit outreach to rural households.

The paper is organized as follows. Section Two details the literature review. Section Three explains the empirical models used to answer the research objectives. This is followed by the data collection procedures, the sampling and survey design. Section Four details the descriptive statistics and regression results for the empirical models. Finally, Section Five summarizes of the main research findings, proposes policy recommendations, gives the limitations of the study and provides suggestions for future research.

## 2. Literature Review

70

#### 2.1. Definitions of Microfinance

There has been a development of a range of terms for defining microfinance in the literature recently. Given the achievement of microfinance as a prominent tool in poverty alleviation, the literature has defined microfinance as follows. Ledgerwood (1998) argued that microfinance is the provision of financial services to low income clients, including small traders, street vendors, small farmers, service providers (e.g. hairdressers, rickshaw drivers), artisans and small producers. Similarly, Mersland and Strøm (2008) state that microfinance provides financial services on a micro scale, such as microcredit, micro insurance, and micro savings for poor and low income people. Meanwhile, Robinson (2002) proposes a broad definition of microfinance, which refers to small scale financial services, primarily credit and savings, provided to people who farm or fish or herd; who operate small enterprises or small business enterprises where goods are produced, recycled, repaired, or sold; provide small services, who work for wages and commissions; gain income from renting agricultural machinery to other individuals and groups at the local level in both rural and urban areas.

In Indonesia, microfinance institutions are regulated by the government. The definition of microfinance institutions was enacted by Law Number 1 Year 2013 which states that microfinance " is a financial institution that is specifically established to provide business development services and community development, either through loans or by financing micro enterprises, to members and the public, the management of deposits, as well as the provision of consulting services for business development, not only profit-oriented but also socially-oriented enterprises". According to the law, MFIs in Indonesia include two types of microfinance, namely: (1) Cooperatives and limited liability companies. The microfinance institutions as cooperative legal institutions are supervised and regulated by the Ministry of Cooperative and Small Medium Enterprises; and (2) other non-bank financial institutions that are governed by the Ministry of Law and Human Rights (see Table 1). Furthermore, the Indonesian Financial Authority (OJK) advises that microcredit providers should focus their microcredit programs or services on the unmet credit demands which, in a large part, have been served by commercial bank or formal Financial Institutions (FIs).

## 2.2. Characteristics and History of Microfinance in Indonesia

The development of Indonesian microfinance can be traced back to the establishment of the *Purwokerto Support and Savings Bank for Netherlands Indies Civil Servants* in 1895 as the origin of Bank Rakyat Indonesia (BRI), the largest microcredit provider in Indonesia. Historically, Indonesian microfinance was initiated to protect poor and indigenous people against the practices of moneylenders and pawn-brokers, carried out mostly by the Chinese and Arabs (Robinson 2002, Revindo & Gan 2017a).

There are two major institutional providers of MFIs services in rural areas: the government-owned Bank Rakyat Indonesia (BRI), with some 3,500 subbranches (*unit desa*) at the sub-district level, and some 9,000 formal and semi-formal MFIs (Seibel & Parhusip 1998). Based on Prawiranata (2013) and the Banking With The Poor (2013), the types of MFIs in Indonesia can be categorized as banks and non-banks (see Table 1).

Indonesian MFIs mostly target rural households as their clients. One of the state-owned banks which predominantly serves micro-lending to rural households in Indonesia is Bank Rakyat Indonesia (BRI).

| Category                                  | Ba  | Iks  | Non-E  | lanks  |
|---|---|--|--|--|
|   | Conventional  | Islamic  | Conventional   | Islamic  |
| State owned banks                         | 1. State-owned Banks:<br>- Based on Act No. 10/1998   | 1. State Owned Banks:<br>- Based on Act No. 10/1998  | 1. Cooperatives:<br>Based on Cooperatives Act No.  | 1. Islamic Cooperatives:<br>Based on Cooperatives Act No.        |
| or  | - Government shares   | - Government shares  | 25/1992<br>Supervised by Ministry of Coopera-<br>tives and MSMF's  | 25/1992<br>Supervised by Ministry of Coopera-<br>tives and MSMFs |
|   | - Supervised by Financial Authority<br>Service (OJK)  | <ul> <li>Supervised by Financial Authority<br/>Service (OJK)</li> </ul>                            | Example:   | Example:   |
| Cooperatives                              | Example:<br>- BRI Units<br>- Mandiri Bank<br>- BTN  | Example:<br>- BRI Syariah<br>- Mandiri Syariah<br>- BTN Syariah                                    | - Koperasi Simpan Pinjam   | - Baitul Maal wa Tamwii (BMT)                                    |
| Commercial Banks                          | 2. Commercial Banks   | <ol> <li>Islamic Commercial Banks</li> <li>Based on Act No. 10/1998</li> </ol>                     | 2. Pawnshops<br>Divided into:  | n/a*   |
| or  | - Based on Act No. 10/1998  | - Private companies  | a. State-owned: Perum Pegadaian (PP)   |  |
|   | - Private companies   | <ul> <li>Supervised by Financial Authority<br/>Service (OJK)</li> </ul>                            | b. Private Pawnshops   |  |
| Non-formal MFIs                           | - Supervised by Financial Authority<br>Service (OJK)  | Example:   |  |  |
|   | Example:<br>- Danamon Simpan Pinjam<br>- BTPN   | - Danamon Syariah<br>- BTPN Syariah  |  |  |
| Rural Banks                               | 3. Rural Banks:   | 3. Islamic Rural Banks:  | 3. Money-lenders, Arisan (self-help  | n/a  |
|   | - Based on Act No. 10/1998  | - Based on Act No. 10/1998   | group):<br>They are not regulated by the gov-<br>ernment and charge very high inter-<br>est rates for borrowers. |  |
| or  | <ul> <li>Private companies</li> <li>Supervised by Financial Authority<br/>Service (OJK)</li> </ul>  | <ul> <li>Private companies</li> <li>Supervised by Financial Authority<br/>Service (OJK)</li> </ul> |  |  |
| Non-formal MFIs                           | <ul> <li>Coverage area: district<br/>level/municipal<br/>Example:</li> </ul>  | - Coverage area: district<br>level/municipal<br>Example:   |  |  |
|   | <ul> <li>Bank Perkreditan Rakyat<br/>(BPR)/Rural Conventional Banks</li> <li>Village Funds and Credit Institu-<br/>tions (Badan Kredit Desa)</li> </ul> | BPR Syariah  |  |  |
| Source: Adapted from * n/a: not available | Prawiranata (2013)  |  |  |  |

Economics and Finance in Indonesia Vol. 65 No. 1, June 2019

SANTOSO, D. B., & GAN, C./MICROCREDIT ACCESSIBILITY IN RURAL HOUSEHOLDS ...

Table 1: Types of Microfinance Institutions (MFIs) in Indonesia

By December 2014, BRI has taken the lead in distributing microcredit to the majority of Indonesia rural households seeking loans with a total outstanding value of IDR24.038 trillion (distributed to more than 11.326 million clients<sup>3</sup>). In other words, BRI has become the largest microcredit provider in Indonesia.

However, there are also several types of MFIs that cater to microcredit for rural households in Indonesia (see Table 1). Those MFIs comprise formal, semi-formal and informal financial institutions that have been operating in the rural areas as providers of microcredit. Arsyad (2005, p. 67) states that a formal financial institution comprises of "a financial institution that is governed by the Indonesian Government, and subject to regulation and supervision by the State"; while the informal financial institution comprises of "intermediaries that operates outside the framework of government regulation and supervision". In addition, the author argued that between the two forms, semiformal MFIs are microcredit providers that are not regulated by banking authorities but are registered and/or licensed by other authorities or regional governments, such as the Ministry of Cooperative and MSMEs and provincial governments (Arsyad 2005).

## 2.3. Microcredit Accessibility in Indonesian Rural Households

Despite the longevity of the existence of microfinance and outreach, Indonesia MFIs have been facing obstacles in catering for microcredit, particularly to rural households who still live under the poverty line. One of the reasons for these obstacles is due to the heterogeneous type of MFIs and the fragmentation and legislation of microcredit institutions, which was also common in other developing countries, especially in Asia (see for example Seibel & Rachmadi 2009, Revindo & Gan 2017b, 2017c, Lebovics, Hermes, & Hudon 2016, Wijesiri, Viganò, & Meoli 2015).

Apart from conventional financial institutions, who act as formal FIs, such as commercial banks (public and private), there are also many semi-formal MFIs which provide microcredit to rural borrowers in Indonesia. These are cooperatives, money-lenders and pawn-shops. These MFIs are supervised and governed by different government agencies. According to Indonesian Banking Laws (No. 7 / 1992 superseded by Law No. 10 / 1998), there are two types of formal banking-institutions serving microfinance. First, there are commercial banks such as BRI Unit, Bank Mandiri, and Bank Pembangunan Daerah (BPD), which cater for microcredit in unit divisions with nationwide coverage area. The second type comprises of rural banks such as Bank Perkreditan Rakyat (BPR) with a major focus on microcredit services to rural households in the district level. Both commercial and rural banks serve microcredit to rural households<sup>4</sup>. On the other hand, cooperatives as semi-formal FIs, are overseen under the supervision of the Ministry of Small Medium and Enterprises (Law No. 25/1992). Furthermore, although Indonesia has been a global leader in microfinance outreach and innovation for the past 25 years, accessibility to microcredit services by the poor is declining (Rosengard & Prasetyantoko 2011).

The lack of financial services to Indonesia rural households has become a major concern to policy makers. In 2009, the World Bank reported in one of their surveys on Indonesia Rural Households Access to financial services that:

 Around half of Indonesian rural households have access to formal financial institutions

<sup>&</sup>lt;sup>3</sup>Data from http://komite-kur.com/article-103-sebaranpenyaluran-kredit-usaha-rakyat-periode-november-2007november-2014.asp.

<sup>&</sup>lt;sup>4</sup>Source: www.bi.go.id.

Economics and Finance in Indonesia Vol. 65 No. 1, June 2019

(more than 50% have savings accounts, especially with commercial banks) while one third do not have access to either formal or informal Fls.

 Fewer than 20% of the Indonesian people have been granted loans from formal banks, with 33% gaining loans from informal financial institutions. Surprisingly, 40% have never had access to credit, with a majority of the population residing in the rural area (World Bank 2009).

This World Bank survey also revealed that the constraints of rural households in accessing finance comprise of a lack of collateral, inadequate identification and documentation, insufficient income and heavy indebtedness.

In addition, Tambunan (2015) found that a lack of capital is the key constraint facing rural households who operate microenterprises. Moreover, the author argues that the lack of capital is mainly due to the lack of access to banks and other formal nonbank financial institutions. Tsukada, Higashikata, and Takahashi (2010), in a study of rural bank practices in East Java Province, have also argued that the poorest households have a relatively lower probability of exploring new credit opportunities than middle-income households, even if the credit scale was very small. Miyata and Sawada (2006) also recognized that credit barriers acted as serious constraints for rural households in adopting new floating net aquaculture technology. In other words, as Miyata and Sawada pointed out, without sufficient capital poor households cannot implement new technology for their income generating activities.

## 3. Method

The binary logistic regression model was chosen to estimate the factors influencing the accessibility of credit for rural households and to determine the characteristics of the clients and non-clients of microcredit.

## 3.1. Identifying Borrowers and Non-borrowers of Microcredit

This is to test the socio-demographic attributes of rural households who did or did not use microcredit. The logit model is used, since the dependent variable  $(Y_{\rm in})$  represents two groups of credit borrowers. The binary outcome of  $(Y_{\rm 1n})$  represents microfinance clients and non-clients of microfinance. The general parametric functional expression of this model can be given as follows:

$$Y_{1n} = \ln\left(\frac{P_i}{1 - P_i}\right) = Z_i \tag{1}$$

where  $\rm Y_{in}$  is credit borrowers (where 1= use microcredit; 0 = do not use microcredit) and  $\rm Z_i$  is a set of explanatory variables and the error terms. Table 2 depicts the explanatory variables and the expected a priori sign used in equation above.

The independent variables comprise of gender, age of consumers, marital status, educational achievement, occupation, household income, expenditure, loan duration, and purpose of loan. The positive and significant coefficients are hypothesized for the consumer's age, marital status, educational level, occupation, household income, loan duration, and purpose of loans. The age of borrowers, educational attainment, household income per month, monthly expenditure by borrowers, and loan duration, as detailed in the previous logit model, are expected to be positive and significant factors in determining clients and non-clients of MFIs (Chaudhary & Ishfaq

| Variable name          | Variable label | Coding   | Expected sign |
|------------------------|----------------|--|---------------|
| Gender of consumers    | GEND           | 1= male; 0=female  | (+/-)         |
| Age of borrowers       | AGE            | 1=18-25 years old; 2=26-35 years old; 3=36-45 years old; 4=46-55   | (+)           |
|                        |                | years old; 5=56-65 years old; 6 = more than 65 years old   |               |
| Marital status         | MAR            | 1=Single/Never Married; 2=Married; 3=Widow/Widower; 4=Di-<br>vorced/Separated  | (+)           |
| Educational attainment | EDU            | 1=Non-formal; 2=Primary; 3=Secondary; 4=High School; 5=College; 6=Bachelor; 7=Postgraduate   | (+)           |
| Occupation             | OCCUP          | 1= Crop farmer; 2= Fisherman; 3= Factory worker; 4= Seasonal worker/casual jobs; 5= Small entrepreneur; 6= Public Servant/<br>Army/police; 7= Retired; 8= unemployed; 9= others  | (+)           |
| Household income       | INCOME         | 1=< IDR 330,776.00; 2=IDR 330,776.00 - IDR 1,163.800.00;<br>3=IDR 1.163,800.00 - 5,000,000.00; 4=> IDR 5,000,000.00  | (+)           |
| Household expenditure  | EXPEND         | 1=< IDR 330,776.00; 2=IDR 330,776.00 - IDR 1,163.800.00;<br>3=IDR 1.163.800.00 - 5,000,000.00; 4=> IDR 5,000,000.00  | (-)           |
| Loan duration          | DUR            | 1 = < 6 months; $2 = 6$ months to 1 year; $3 = 1$ to 2 years; $4 = 2$ to 3 years; $5 = > 3$ years  | (+)           |
| Purpose of loan        | PURPOSE        | 1=Expanding business, manufacturing, trading, or service activi-<br>ties; 2=Buying car / motorcycle; 3=Emergency needs; 4=Social<br>needs; 5=Daily consumptions; 6=Financing new small project(s);<br>7=Paying other debts; 8=Others | (+)           |

Table 2: Summary of Explanatory Variables used in Binary Logit Model

2003, Evans et al. 1999).

Evans et al. (1999) also indicated that there is a negative and significant correlation between household expenditure and the clients or non-clients of microcredit. This is because households who spent more of their monthly expenditure would be less likely to participate in microcredit. Further, Chaudhary and Ishfaq (2003) and Li, Gan, and Hu (2011a) suggest that the occupation of clients could also have an influence on the factors that differentiate between the clients of microcredit. A positive and significant determinant is expected because clients who engage in stable and productive income-generating activities are more likely to become clients of microcredit. Moreover, Sebopetji and Belete (2009) found that the marital status of rural households affected their likelihood of being microcredit clients. However, the gender impact of household borrowers is indeterminate. Churchill (1999) and Li, Gan, and Hu (2011c) studies have shown a positive and significant correlation for the likelihood of rural borrowers and female participation in microcredit. Further, Evans et al. (1999) have indicated that other risk factors, such as low education, small family size, and being landless, are negative and significant factors influencing the probability of a rural household to use microcredit in Bangladesh. The findings imply that having lower education, small household size, lower assets reduce the probability of microcredit participation. Finally, purpose of loan is hypothesized as a positive and significant determinant affecting the probability of rural households being clients of MFIs. Ho (2004) has concluded that microcredit used for consumption purpose positively and significantly affects rural household being borrowers of informal MFIs.

## 3.2. Determinants of Credit Accessibility

There is no direct method to measure credit accessibility. However, it can be examined indirectly by using an empirical study of formal and informal borrowings. Karugia et al. (2005) and Ravi (2004) used formal and informal borrowings as an indicator of credit accessibility. Both studies have used logistic models to test the determinants of the likelihood of whether rural households have been able to access microcredit (credit granted) or were re-

jected. This current study also used the borrowings observed from formal, informal, and other microcredit lenders as a proxy for credit access. Further, this study measured credit accessibility of rural households, whether or not their credit applications were approved. Previous studies (see for instance, Kashuliza & Kydd 1996, Li, Gan, & Hu 2011a, Umoh 2006, Vaessen 2001) that determined rural households' accessibility to microfinance took into account both household and institutional level characteristics. This current study follows the Raleting and Obi (2015) binary logistic regression model to analyze the accessibility of microcredit from the perspective of the rural household. Binary logistic regression is useful when the dependent variable is dichotomous (Chan 2005). The binary model has been used widely in a number of fields including the social sciences when investigating dichotomous variables (Mohamed 2003, Scott Long 1997). Using the logit regression model, Vaessen (2001) determined the factors affecting microcredit accessibility in Nicaragua. Vaessen tested the probability of rural households influenced by observable factors. Those variables include education, age, family size, household assets, collateral, type of business and networking. The binary results of the logistic model measures whether rural households' loan application was accepted or rejected (Chaudhary & Ishfaq 2003, Li, Gan, & Hu 2011a, Umoh 2006). Thus, the binary response in our study defines Y=1 for loan application accepted and Y=0 for loan application rejected. The parametric functional form can also be written as follows:

$$Y_{1n}^{*} = \ln\left(\frac{P_{i}}{1 - P_{i}}\right) = Z_{i}$$
 (2)

Where  $Y_{in}^*$  = Decision on microcredit application, (where 1= approved and 0= rejected);  $P_{in}$  = the probability of choices and  $Z_i$  is a vector of explanatory variables and the error terms. Table 3 depicts the explanatory variables used in equation 7 and the expected a priori sign of the variables.

Previous empirical studies by Li, Gan, and Hu (2011a), Sarap (1990), Tang, Guang, Z., and Jin (2010), Vaessen (2001), and Wydick, Hayes, and Kempf (2011) investigated the household characteristics that influenced credit accessibility. These household characteristics include household size, networking of clients, educational attainment, income, household assets, collateral and interest rates. The studies have found positive indeterminate and negative significant determinants that influenced the probability of microcredit access. For example, age and distance to the nearest MFIs were indeterminate factors that affected microcredit accessibility. Li, Gan, and Hu (2011a) showed the distance variable negatively influenced the likelihood of rural households in China to access microcredit. The study found that households that resided more than 20 lis (10 km) from the nearest MFIs branch were less likely to be granted credit because of higher transaction and time opportunity costs.

Further, the age of the borrower was a key factor affecting borrowers' access to microcredit. Mohamed (2003) found that the age of the clients had a negative but significant relationship with credit acceptance. This finding suggested that older people were less likely to get credit approval than younger borrowers. In addition, Ho (2004), Li, Gan, and Hu (2011a), and Wydick, Hayes, and Kempf (2011), found that the assets of borrowers have a negative and significant relationship with credit accessibility. This means that households who are wealthier or less budget constrained are less likely to apply for microcredit. In contrast, income of borrowers also exhibits a positive correlation with loan acceptance, meaning that the higher the income of the rural household borrowers, the greater the probability that the borrower's loan application will be accepted (Evans et al. 1999, Li, Gan, & Hu 2011a, Mohamed 2003).

| Variables Name                | Variable label | Coding  | Expected signs |
|-------------------------------|----------------|---|----------------|
| Gender of clients             | GEND           | 1=male and 0=female   | (+/-)          |
| Age of borrowers              | AGE            | 1=18-25 years; 2=26-35 years; 3=36-45 years; 4=46-55 years; 5=56-65 years; 6= above 65 years                        | (+/-)          |
| Household size                | HSIZE          | 1 =1 member; 2=2 members; 3=3 members; 4=more than 3 mem-<br>bers   | (+)            |
| Networking                    | NET            | 1= with networking; 0=otherwise   | (+)            |
| Educational attainment        | EDU            | 1=Non-formal; 2=Primary; 3=Secondary; 4=High School, 5=College; 6=Bachelor; 7=Postgraduate                          | (+)            |
| Household income<br>(monthly) | INCOME         | 1=< IDR 330,776.00; 2=IDR 330,776.00 - IDR 1,163.800.00;<br>3=IDR 1.163,800.00 - 5,000,000.00; 4=> IDR 5,000,000.00 | (+)            |
| Household assets              | ASSET          | 1=Farm lands / buildings; 2=Livestock; 3= Agricultural machiner-<br>ies; 4= Car / Motorcycles; 5=Others             | (-)            |
| Collateral                    | COL            | 1= with collateral, 0=otherwise   | (+)            |
| Distance to MFIs              | DIST           | 1=< 5Km; 2=5 – 10 Km; 3=10-20 Km; 4 = >20 Km  | (-)            |
| Loan Duration                 | DUR            | 1= less than 6 months; 2=6 months - 1 year; 3=1 - 2 years; 4=2-3 years; 5(more than 3 years)                        | (+)            |

Table 3: Summary of Explanatory Variables in Binary Logit Model

The impact of *gender* on credit accessibility was indeterminate. Okten and Osili (2004) found that females were more likely to be granted credit. The authors indicated that microcredit providers preferred lending to females due to their lower default rates. However, female borrowers were also found to be discriminated against in the credit market (Ho 2004, Kashuliza & Kydd 1996, Zeller 1994).

76

A positive and significant correlation between credit accessibility and other explanatory variables could be found for household size, recommendations, educational attainment, collateral, and loan duration. Studies in South Africa (Okurut 2006) and India (Sarap 1990), have shown that the larger the family size, the more likely they are to be able to access microcredit. Recommendations, as a proxy for networking to MFIs, also played a significant role in influencing microcredit approval (see for example, Coleman 2006, Okten & Osili 2004, Vaessen 2001). This implies that rural borrowers with closed relationship with MFIs officers would have a higher probability of accessing microcredit. Moreover, there are positive and significant correlations between level of education and credit accessibility for rural households. Evans et al. (1999) revealed that rural households with higher levels of educational attainment are more likely to take advantage of the microcredit market. The collateral variable is hypothesized to positively influence credit accessibility. A study by Nagarajan and Meyer (1996) found that by providing collateral, rural borrowers would increase their probability of accessing microcredit. Further, there is a positive and significant correlation between *loan duration* and credit accessibility. Abaru et al. (2006) found that loan duration positively affects the probability of rural farmers' access to agricultural credit in Uganda.

#### 3.3. Data Collection Procedure

#### 3.3.1. Sampling Design

The study sample size is determined by the Cochran (2007) formula used in most primary data collection, as follows:

$$n = \frac{z^2 pq}{e^2}$$
(3)

Where,  $\rm n$  is sample size,  $\rm z^2$  is the square of the critical value of the normal curve that cuts off an area defining significance at the tails,  $\rm e$  is the desired level of precision,  $\rm p$  is the estimated proportion of an attribute that is present in the population, and  $\rm q$  is  $1-\rm p.$  This study used the 95% (or  $\pm 5\%$  pre-

cision) level of confidence and assumed p = 0.5and q = 0.5. Therefore, based on equation 3.12, the total minimum sample size for our study is 385 rural households in Indonesia. The current study interviewed 591 respondents to obtain sufficient completed responses for the analysis. The survey was conducted in the Special Province of Yogyakarta, in particular, the Bantul district, from February-March 2016. The Bantul is located in Yogyakarta Special Region in Indonesia. It lies between 07° 44' 04" - 08° 00' 27" south latitude and 110° 12' 34" -110° 31' 08" east longitude and has a population of 919,440. Amongst Bantul residents (25.56%) still relied on the agricultural sector as their main occupation; while 21.16% of them worked in trades and the other 19% and 17% of people worked in small industries and the services, respectively<sup>5</sup>. Among those residents, there are 44,778 micro, small and medium enterprises (MSMEs) which could potentially demand microcredit. Meanwhile, there are 19 formal MFIs that catered for small loans (including commercial and rural banks/BPR). In addition, the Industry and Cooperative District Office (Disperdagkop) estimates there are 463 informal MFIs (cooperatives)<sup>6</sup>.

#### 3.3.2. Survey Instruments

In order to address the research objectives of this study, a structured questionnaire was developed to obtain the data for empirical analysis. The questionnaire was submitted to the Lincoln University Human Ethics Committee for approval. The questionnaire was printed bilingually, in English and *Bahasa Indonesia*. The translation into Bahasa Indonesia was verified by asking several Indonesian students to read the questionnaire and giving the feedback on the draft. Before the administering the survey questionnaire, the questionnaire was randomly pilottested with 20 rural households in Bantul District. This helped to rectify any ambiguities before administering the questionnaire to the sample rural households.

The survey questions are based on the literature and the overall objective of the study. The structured questionnaire consisted of three sections: Section 1 identified the determinants of credit accessibility for Indonesian rural households. The questions measured information on respondents' sources of finance, the amount and purpose of their loans, as well as the duration of them, the interest rates and interest repayment methods. Section 2 of the questionnaire focused on the welfare impacts of microfinance (microcredit) on rural households in Indonesia. In this regard, the socio-demographics factors of the households and other household characteristics, such as household networking, wealth and assets, and village/commune characteristics factors, such as distance to nearest to MFI, are tested for significance of the covariates on the households' welfare. Section 3 includes the profile of the rural households' demographic characteristics such as age, ethnic, marital status, household characteristics, educational level, and experience.

#### 3.3.3. Data Collection Process

The field-work for collecting the primary data was conducted over two months; February and March 2016. Nine research assistants helped to interview the participants during the survey. Rural households were asked if they would voluntarily agree to participate in the research and if they responded with a yes, survey assistants would return to collect the completed questionnaires. A total of 591 survey questionnaires were administered and returned, of which 488 responses were usable, generating a response rate of 82.57%. Completed questionnaires

<sup>&</sup>lt;sup>5</sup>Data from: https://bantulkab.go.id/datapokok/1001\_ lembaga\_keuangan.html.

<sup>&</sup>lt;sup>6</sup>Data from: http://perindagkop.bantulkab.go.id/.

were then coded, entered into Excel, and imported into STATA 13 software for analysis.

## 4. Results

#### 4.1. Profile of the Respondents

Table 4 shows the differences between microcredit borrowers and non-borrowers in terms of individual characteristics (gender, age, marital status, educational level, occupation, and work experience) and household characteristics (number of income earners, number of household members, household income and household expenditure per month). A chi-square test was performed to assess whether there are significant relationships between credit acceptance and the respondents' characteristics.

## 4.2. Empirical Results Related to Clients and Non-Clients of Microcredit

Overall, the binary logit model correctly predicts 86.27% of the respondents' client and non-client status. The likelihood ratio test with a chi-square of 19.06 (p < 0.05) leads to rejection of the null hypothesis that all variable coefficients are equal to zero. Therefore, the binary model can be used to explain the determinants differentiating clients and non-clients of MFIs in the surveyed area.

The results displayed in Table 5 show that the gender of respondents did not significantly determine the probability of rural households becoming clients of MFIs. This finding is similar to a study in Ghana by Ayamga et al. (2006), but contradict the studies of Okten and Osili (2004), and Sebopetji and Belete (2009), who found that gender significantly affects rural households' participation in microfinance programs. Ayamga et al. (2006) showed that the gender of borrowers is not a significant factor influencing microcredit participation in Northern Ghana, even though the a priori expectation was significant. An explanation for this is that in Northern Ghana, males usually control household resources such as land, labor and even farm output. Therefore, women borrowers in Ghana were usually credit constrained (Ayamga et al. 2006). On the other hand, Okten and Osili (2004)'s study found that female borrowers significantly affect the probability of microcredit participation in Indonesia. Similarly, Chaudhary and Ishfaq (2003) have argued that female borrowers are more likely to participate in microcredit because they have more reliable repayment behavior than male counterparts in Bangladesh.

As hypothesized, the age of household is a significant factor affecting the likelihood of being clients of microcredit (z = 1.72 and p = 0.10). This result implies that older rural households are more likely to borrow microcredit than younger households. The result supports earlier empirical results (see Anggraeni 2009, Wydick, Hayes, and Kempf 2011). Anggraeni (2009) found that the age of borrowers is a key factor in determining clients of Rotating Savings and Credit Association in West Java, Indonesia, while Wydick, Hayes, and Kempf (2011) argue that the age variable significantly affects rural households in Guatemala participation in microcredit programs. The results in Table 5 also show the marginal effect of age variable as 0.029. This means a one-unit increase in a client's age group increase the probability of being clients of MFIs by 2.9%.

In terms of marital status, Table 5 shows that the coefficient of marital variable is statistically significant and affects the probability of the household decision in applying for microcredit (z = 2.01 and p = 0.044). This result can be interpreted to mean that rural households who are married are more likely to use microcredit. This result confirmed the study in South Africa by Sebopetji and Belete (2009), which found

| Characteristics                       | Non-Applicants |              | Applicants      |                 | All respondents $(n - 488)$ |             | Statistical Test                    |
|---------------------------------------|----------------|--------------|-----------------|-----------------|-----------------------------|-------------|-------------------------------------|
| of the Respondents                    | (n             | $1_1 = 97$ ) | (n <sub>2</sub> | = 391)          | (n :                        | = 488)      |                                     |
| Individual Observatoriation           | n              | % Of $n_1$   | n               | % Of ${ m n}_2$ | n                           | % of n      |                                     |
| Individual Characteristics            |                |              |                 |                 |                             |             |                                     |
| Female                                | 62             | 63.9         | 245             | 627             | 307                         | 62 9        | $X^2 = 0.053$                       |
| Male                                  | 35             | 36.1         | 146             | 37.3            | 181                         | 37.1        | p=0.05                              |
| Age group                             |                |              |                 |                 |                             |             |                                     |
| 18–25 year-olds                       | 11             | 11.3         | 16              | 4.1             | 27                          | 5.5         | X <sup>2</sup> = 13.77**            |
| 26–35 year-olds                       | 32             | 33           | 100             | 25.6            | 132                         | 27          | p=0.05                              |
| 36–45 year-olds                       | 28             | 28.9         | 152             | 38.9            | 180                         | 36.9        |                                     |
| 46–55 year-olds                       | 16             | 16.5         | 84              | 21.5            | 100                         | 20.5        |                                     |
| 56–65 year-olds                       | 7              | 7.2          | 34              | 8.7             | 41                          | 8.4         |                                     |
| Over 65 year-olds                     | 3              | 3.1          | 5               | 1.3             | 8                           | 1.6         |                                     |
| Marital status                        | 10             | 10.0         | 00              | 7.0             | 40                          | 0.4         | V2 14 05***                         |
| Single/Never Married                  | 18             | 18.6         | 28              | 7.2             | 40                          | 9.4         | $X^{-}=14.05^{-10}$                 |
| Widow / Widower                       | 73             | 75.5         | 10              | 26              | 421                         | 00.3<br>2 7 | p=0.01                              |
| Divorced/Separated                    | 3              | 3.1          | 5               | 1.3             | 8                           | 1.6         |                                     |
| Educational level                     |                | 0.11         |                 |                 |                             |             |                                     |
| No formal education                   | 3              | 3.1          | 7               | 1.8             | 10                          | 2           | X <sup>2</sup> =10.15               |
| Primary School                        | 4              | 4.1          | 42              | 10.7            | 46                          | 9.4         | p=0.05                              |
| Secondary/Junior High School          | 13             | 13.4         | 74              | 18.9            | 87                          | 17.8        | •                                   |
| High School                           | 51             | 52.6         | 178             | 45.5            | 229                         | 46.9        |                                     |
| College / Vocational                  | 9              | 9.3          | 39              | 10              | 48                          | 9.8         |                                     |
| Bachelor                              | 14             | 14.4         | 48              | 12.3            | 62                          | 12.7        |                                     |
| Postgraduate                          | 3              | 3.1          | 3               | 0.8             | 6                           | 1.2         |                                     |
| Occupation                            |                |              |                 |                 |                             |             | 77 <sup>2</sup> (0.40 <sup>++</sup> |
| Crop farmer                           | 10             | 10.3         | 46              | 11.8            | 56                          | 11.5        | X <sup>2</sup> =18.43**             |
| Fisherman                             | 2              | 2.1          | 4               | 1               | 6                           | 1.2         | p=0.05                              |
| Factory worker                        | 15             | 15.5         | 63              | 16.1            | /8                          | 16          |                                     |
| Small entrepreneur                    | 32             | 33           | 20              | 1.2             | 213                         | 13.6        |                                     |
| Public Servant / Army / police        | 10             | 10.3         | 33              | 84              | 43                          | 8.8         |                                     |
| Retired                               | 8              | 82           | 14              | 3.6             | 22                          | 4.5         |                                     |
| Unemployed                            | 2              | 2.1          | 15              | 3.8             | 27                          | 5.5         |                                     |
| Others                                | 12             | 12.4         | 15              | 3.8             | 27                          | 5.5         |                                     |
| Working duration                      |                |              |                 |                 |                             |             |                                     |
| Less than 1 year                      | 11             | 11.3         | 25              | 6.4             | 36                          | 7.4         | $X^2 = 6.47$                        |
| Between 1 and 3 years                 | 24             | 24.7         | 69              | 17.6            | 93                          | 19.1        | p=0.05                              |
| Between 3 and 5 years                 | 21             | 21.6         | 90              | 23              | 111                         | 22.7        |                                     |
| Between 5 and 10 years                | 13             | 13.4         | 69              | 17.6            | 82                          | 16.8        |                                     |
| More than 10 years                    | 28             | 28.9         | 138             | 35.3            | 166                         | 34          |                                     |
| Household Characteristics             |                |              |                 |                 |                             |             |                                     |
| Number of income earners in household | ~~             |              | 110             | <u> </u>        | 1.10                        | 00.0        | w <sup>2</sup> + 044                |
|                                       | 33             | 34           | 113             | 28.9            | 146                         | 29.9        | X <sup>2</sup> =1.044               |
| 2 earners<br>More then 2 earners      | 48             | 49.5         | 213             | 54.5            | 261                         | 53.5        | p=0.05                              |
| Number of household members           | 10             | 10.5         | 65              | 10.0            | 01                          | 10.0        |                                     |
| 1 member                              | 19             | 19.6         | 84              | 21.5            | 103                         | 21.1        | X <sup>2</sup> -0 29                |
| 2 members                             | 28             | 28.9         | 105             | 26.9            | 133                         | 27.3        | p=0.05                              |
| 3 members                             | 28             | 28.9         | 110             | 28.1            | 133                         | 27.3        |                                     |
| More than 3                           | 22             | 22.7         | 92              | 23.5            | 114                         | 23.4        |                                     |
| Household income (per month)          |                |              |                 |                 |                             |             |                                     |
| < IDR 330,776.00                      | 6              | 6.2          | 22              | 5.6             | 28                          | 5.7         | X <sup>2</sup> =3.47                |
| IDR 330,777.00 - IDR 1,163,800.00     | 47             | 48.5         | 155             | 39.6            | 202                         | 41.4        | p=0.05                              |
| IDR 1,163,801.00 - IDR 5.000.00.00    | 37             | 38.1         | 190             | 48.6            | 227                         | 46.5        |                                     |
| > IDR 5.000.001.00                    | 7              | 7.2          | 24              | 6.1             | 31                          | 6.4         |                                     |
| Household expenditures (per month)    | ~              |              |                 |                 |                             | <b>.</b> .  | **?                                 |
| < IDK 330,776.00                      | 9              | 9.3          | 22              | 5.6             | 31                          | 6.4         | X <sup>2</sup> =3.06                |
| IDR 330,///.00 - IDR 1,163,800.00     | 51             | 52.6         | 198             | 50.6            | 249                         | 51          | p=0.05                              |
| DR 1,163,801.00 - IDK 5.000.00.00     | 33             | 34<br>4 1    | 160             | 40.9            | 193                         | 39.5        |                                     |
| 7 IDN 3.000.001.00                    | 4              | 4.1          | 201             | 2.0<br>100      | 10                          | 100         |                                     |
| ισιαι                                 | 31             | 100          | 031             | 100             | 1 400                       | 100         |                                     |

#### Table 4: Profile of the Surveyed Respondents (Individual and Household Characteristics)

 Total
 97
 100
 391
 100
 488

 Note: 1. \*\*\*; \*\*; \* indicates the significance levels at 1%, 5%, and 10% respectively

Table 5: Results of Logistic Regression on Clients/Non-Clients of Microcredit

| Number of observations               | 488         |           |       |       |                                 |
|--------------------------------------|-------------|-----------|-------|-------|---------------------------------|
| Log likelihood                       | -185.681    |           |       |       |                                 |
| LR chi <sup>2</sup> (9)              | 19.06       |           |       |       |                                 |
| Prob. > $chi^2$                      | 0.0247      |           |       |       |                                 |
| McFadden Pseudo R <sup>2</sup>       | 0.0488      |           |       |       |                                 |
| Degrees of freedom                   | 8           |           |       |       |                                 |
| % Correctly predicted                | 86.27%      |           |       |       |                                 |
| Independent Variables <sup>(1)</sup> | Coefficient | Std. Err. | Z     | p> z  | Marginal Effects <sup>(2)</sup> |
| GEND                                 | 0.177       | 0.296     | 0.60  | 0.549 | 0.020                           |
| AGE                                  | 0.252*      | 0.146     | 1.72  | 0.085 | 0.029                           |
| MAR                                  | 0.757**     | 0.376     | 2.01  | 0.044 | 0.086                           |
| EDU                                  | -0.245**    | 0.122     | -2.00 | 0.045 | -0.028                          |
| OCCUP                                | -0.004      | 0.076     | -0.05 | 0.959 | 0.000                           |
| INCOME                               | 0.387       | 0.279     | 1.39  | 0.166 | 0.045                           |
| EXPEND                               | -0.248      | 0.298     | -0.83 | 0.407 | -0.020                          |
| DUR                                  | -0.004      | 0.151     | -0.03 | 0.979 | 0.000                           |
| PURPOSE                              | -0.073      | 0.087     | -0.83 | 0.404 | -0.008                          |
| _cons                                | 0.390       | 0.896     | 0.44  | 0.663 |                                 |

Note: (1) Dependent Variable: 1 if households are clients of MFIs and 0 otherwise (2) Marginal Effects are in the mean value

, \*\*,\* indicate the significance levels at 1%, 5%, and 10%, respectively

that married borrowers are more likely to participate in the microcredit program. In contrast, Nouman et al. (2013) pointed out that the probability of being microcredit clients in Pakistan is affected by the marital status of borrowers. The authors revealed a negative and significant coefficient for marital status, which implies that married farmers are less likely to get a larger amount of credit compared to unmarried farmers. One possible reason for this is that married farmers are likely to have more stable incomes and thus demand less microcredit than unmarried farmers. Table 5 also shows the marginal effect of marital status as 0.086. This result indicates that a one-unit change in marital status, would lead to an increase in the probability of being a microfinance borrower by 8.6%.

Educational attainment is also a key factor discriminating between clients or non-clients of microcredit. As shown in Table 5, educational level significantly influences rural households in becoming credit borrowers (z = -2.00 and p = 0.05). The result can be interpreted to mean that the higher the educational level of households/respondents, the less likely they will use microcredit. One possible reason is that more highly educated households can easily

access formal financial institutions sources which offer lower interest rates. This result confirmed similar empirical results by Ayamga et al. (2006) in Ghana, which found that the level of secondary education significantly influenced the probability of farmers decision to participate in the microfinance programs. Similarly, Vaessen (2001) argues that the probability of using microcredit is determined by the level of formal education in Northern Nicaragua. Table 5's results also display a marginal effect for education as -0.028. This means that an increase in educational attainment would decrease the probability of rural household becoming a microcredit borrower by 2.8%.

#### 4.3. Empirical Results Related to Credit Accessibility

Among the 390 respondents, 362 households' (92.8%) credit applications were granted, while 28 respondents (7.2%) were not granted. Logistic regression analysis is used to identify the household factors influencing credit accessibility. This empirical approach is used to capture the binary outcome

of loan applications: accepted or rejected by the microfinance institution (coded as 1=accepted or 0=otherwise). Overall, the model correctly predicts 92.82% of credit accessibility. The likelihood ratio test exhibits a significant (p < 0.01) chi-square of presumably 20.48 (pseudo  $R^2 = 0.1017$ , significant p < 0.05) and thus allows rejection of the null hypothesis that all variable coefficients in the logistic model are equal to zero. Hence, the model can be used to explain the factors affecting credit accessibility.

Table 6 shows four variables significantly (p < 0.10) affect credit accessibility: age, income, interest rate and loan duration. Table 6 also depicts the marginal effects of each explanatory variable. Greene (2003) and Train (1986) advocate that the estimated logistic regression coefficient results obtained by maximum likelihood did not yield a direct interpretation, only the sign of each coefficient of the effect of independent variables. As a result, the marginal effects are used to predict the change of the predicted probability associated with the explanatory variables.

Age of borrower positively affects the probability of microcredit access (z = 1.78, significant at p < 0.10). In this current study, we classified the range of ages between 18–25 years coded as (1); 26–35 years as (2), 36–45 years as (3), 46–55 years as (4), 56–65 years as (5) and (6) for the respondents whose ages are above 65 years. This result supports the empirical findings of Li, Gan, and Hu (2011a) who reveal that the age of borrowers affects the credit accessibility of rural households in China. Table 6 also shows a marginal effect of age at 0.019. This result indicates that as respondents' age increases by one unit in the age group, the probability to access microcredit would increase by 1.9%.

Household income positively affects credit accessibility (z = 2.07, significant at p < 0.05) which

indicates that rural households with higher monthly incomes are more likely to access microcredit. A possible reason for this is that higher-income households exhibit more capability for loan repayments, leading to the probability of being microcredit borrowers. This result corresponds to other empirical studies (see for example, Evans et al. 1999, Li, Gan, & Hu 2011a, Mohamed 2003). In addition, Table 6 reveals the marginal effect of monthly income at 0.0309. This means that as the borrowers move up one level in monthly income group the probability of credit approval would increase by 3.1%.

81

Table 6 also shows the positive and significant effect of *interest rates* on credit accessibility (z = 1.74, significant at p < 0.10). This means that microcredit providers used interest rates to screen rural households as their borrowers. This is because rural borrowers who accept higher interest rates are potentially risk-taking borrowers. In other words, there is an adverse selection problem in credit accessibility of rural households in Bantul. This result also corresponds to the study by Gray (2006) who found that interest rates statistically and significantly affect the credit access to rural borrowers. Further, the marginal effect of interest rates is 0.0029, which indicates that rural households who accept 1% higher interest rates exhibit a higher probability of obtaining credit by 0.3%.

In contrast, Table 6 also reveals a negative and significant effect of *loan duration* on credit access (z = -2.38, significant at p < 0.05). This result implies that respondents who applied for shorter loan durations are more likely to be granted loans by the microfinance lenders. One of the potential explanation for this is that MFIs considered the unexpected risks perceived as possible during the microcredit term faced by rural borrowers. The marginal effect of loan duration is -0.026. This implies that as the borrowers apply for an additional unit of loan duration term, the probability of being accepted would decrease by 2.6%. This result confirmed the previ-

| No. of observations                  | 390                   |           |       |       |                                |
|--------------------------------------|-----------------------|-----------|-------|-------|--------------------------------|
| Log likelihood                       | -90.477816            |           |       |       |                                |
| LR chi <sup>2</sup> (11)             | 20.48                 |           |       |       |                                |
| Prob. > $chi^2$                      | 0.0391                |           |       |       |                                |
| Pseudo R <sup>2</sup>                | 0.1017                |           |       |       |                                |
| Correctly predicted                  | 93.08%                |           |       |       |                                |
| Independent Variables <sup>(1)</sup> | Estimated Coefficient | Std. Err. | Z     | p> z  | Marginal Effect <sup>(2)</sup> |
| GENDER                               | 0.358                 | 0.4705    | 0.76  | 0.447 | 0.0176                         |
| AGE                                  | 0.3836*               | 0.2153    | 1.78  | 0.075 | 0.0188                         |
| HSIZE                                | 0.1969                | 0.1991    | 0.99  | 0.323 | 0.0097                         |
| NET                                  | 0.1362                | 0.4312    | 0.32  | 0.752 | 0.0067                         |
| EDU                                  | -0.2464               | 0.1757    | -1.4  | 0.161 | -0.0121                        |
| INCOME                               | 0.6303**              | 0.3044    | 2.07  | 0.038 | 0.0309                         |
| ASSET                                | 0.2297                | 0.1593    | 1.44  | 0.149 | 0.0113                         |
| INTR                                 | 0.0582*               | 0.0335    | 1.74  | 0.082 | 0.0029                         |
| COL                                  | 0.6601                | 0.5006    | 1.32  | 0.187 | 0.0324                         |
| DIST                                 | 0.0031                | 0.3141    | 0.01  | 0.992 | 0.0002                         |
| DUR                                  | -0.5365**             | 0.2253    | -2.38 | 0.017 | -0.0263                        |
| _cons                                | -0.0414               | 14.222    | -0.03 | 0.977 |                                |

Table 6: Results of Binary Logistic Regression on Credit Accessibility

Note: (1) Dependent Variable: 1 if households loan applications are accepted and 0 otherwise (2) Marginal effects are in the mean value

\*\*\*, \*\*,\* indicate the significance levels at 1%, 5%, and 10%, respectively

ous study in the Philippines by Gray (2006). In the empirical study of farmers and fisher-folk's credit accessibility, Gray argued that, formal MFIs tend to provide shorter-term loans because of the seasonality of agricultural activities (in order to minimize credit risk). This is due to the fact that with longer loan terms, credit lenders would be exposed to a higher default risk of microcredit.

With regard to the gender of respondents, the results show a z statistic value of 0.76 and a p-value of 0.447. This means that the gender variable did not influence credit accessibility. This result implies that the gender of rural households did not affect the likelihood of becoming a microcredit borrower. A possible reason for this is that the gender of borrowers was not seen to determine the client's repayment capacity. Further, since microfinance has been introduced, the participation of Indonesian women in microcredit programs has increased, particularly in family decision making and education of their children (Panjaitan-Drioadisuryo & Cloud 1999). In this regard, male and female borrowers have the same opportunity to access microcredit in Indonesia. This finding contradicts Kashuliza and

Kydd (1996) research, who showed a significant influence of the gender determinant in credit accessibility in Tanzania, especially in women's participation of microcredit.

Table 6 shows the z statistics and p-values for *household* size as 0.323 and 0.99, respectively. These results imply that MFIs did not consider household size as a key determinant when selecting their borrowers. Most of the microcredit lenders did not consider the number of family members as one of the credit approval criteria. On possible reason is that MFIs did not screen rural households based on the number of borrowers family member (household size). Moreover, our empirical results contradict the finding of Li, Gan, and Hu (2011a), who found a significant correlation between household size and the likelihood of being microcredit borrowers.

*Networking* did not affect credit accessibility in the surveyed households. The empirical results reveal a non-significant, but positive, coefficient of networking variables (z = 0.32 and p = 0.752). This means that the networking of respondents, such as relationships with credit officers, and local leader

recommendations, did not affect the probability of accessing microcredit. One possible explanation for this is that microcredit providers opt to approve microloans as long as the borrowers meet the terms and conditions required by the MFIs. These results contradict previous research findings (see Coleman 2006, Okten & Osili 2004).

Table 6 shows that *educational attainment* is not a key factor influencing credit access for the surveyed households. The coefficient estimation result exhibits a negative and insignificant sign (z = -1.4and p = 0.161). This result suggests that level of education did not affect rural household's participation in the microcredit market. One possible explanation for this is that with higher education attainment, rural households in Bantul can apply for more stable employment, leading to higher monthly incomes. As a result, more highly educated households who earn greater income opted not to borrow from MFIs.

There is a non-significant, but positive correlation between household assets and accessibility to credit. Table 6 shows that the z statistics and pvalues of household assets were 0.149 and 1.44. The results reveal that the household assets of borrowers did not influence microcredit providers in approving loan applications. This suggests that assets of borrowers do not represent borrower's capability in credit repayment which is considered as determinant in gaining a credit provider's approval, even though assets of borrowers can be used as collateral substitutes for credit lenders (Mohamed 2003). This finding contradicts Li, Gan, and Hu's. (2011a), who state that the assets of borrowers were less likely to influence credit accessibility in rural households in China.

In terms of the *collateral* variable, our empirical results show the z statistic value and p-value as 0.187 and 1.32. In our analysis, we coded respondents who provided collateral, as 1, and without collateral, as 0. The results show a positive but non-

significant correlation between collateral and credit acceptance. This result suggests that collateral is not a key factor influencing credit accessibility. One possible reason is that the surveyed respondents accessed an alternative "group lending" scheme which did not require collateral, such as Program Nasional Pemberdayaan Masyarakat (PNPM) program. This national program in poverty reduction was officially launched in 2007. One of its initiatives is provision of capital and financial resources through revolving funds and microcredit for the poor in order to help the local economy. This program is sponsored by the World Bank and implemented by the Ministry of Home Affairs<sup>7</sup>. This finding differs from Nagarajan and Meyer (1996)'s study in the Philippines, which suggested a correlation between collateral determinants and accessibility of microcredit.

In regard to the distance variable, the regression result shows no significant correlations between distance to nearest MFIs of respondents to accessibility to microcredit (z = 0.01 and p = 0.992). This finding implies that credit providers did not discriminate against borrowers based on their location. One possible reason for this is that there are numerous MFIs in the Bantul district providing microcredit for rural households which causes intense competition amongst credit lenders. As a result, MFIs did not take into account the borrowers' residency for credit approval. Likewise, rural households did not consider distance to MFIs as long as their loan application was approved. This is because transportation in Bantul became relatively easy since every household was likely to have a motorcycle and there has been substantial roads development in recent years. However, this result contradicts a study in Ghana, which found that distance to the nearest MFIs significantly and positively influenced the credit accessibility of rural households (Ayamga

<sup>&</sup>lt;sup>7</sup>Source: http://www.worldbank.org/en/country/indonesia/ brief/community-driven-development-in-indonesia.

Economics and Finance in Indonesia Vol. 65 No. 1, June 2019

et al. 2006).

## 5. Conclusion

This study emerged from the fact that rural families in Indonesia are still lacking in accessing microfinance institutions. Two binary logistic models are used. In the credit accessibility model, the binary logit regression addressed factors influencing credit accessibility in Indonesia. The regression results revealed that the age of borrowers (AGE), monthly income (INCOME), interest rates (INTR) and loan duration (DUR) are key variables that significantly affected the credit accessibility of rural households in Bantul.

The most obvious finding emerging from this model is that monthly income is a key determinant affecting the likelihood of being accepted for microcredit. The results suggest that rural households who increased their monthly income would have a higher probability (by 3.1%) of getting approval for their credit application. In addition, the logistic model shows that older rural householders who applied for microcredit are more likely to have their credit approved. This is likely because older borrowers are more experienced and wiser about utilizing their credit in income generating activities which make them preferable to credit lenders. On the other hand, there are a considerable number of younger nonborrowers in the surveyed area. This means that there are opportunities for microfinance institutions to extend microcredit to younger borrowers.

In terms of interest rates, the model exhibits a positive and significant result, indicating that rural households who accepted higher interest rates have greater chances of credit acceptance, albeit the effect is moderate. In contrast, loan duration is found to be a negative and significant variable influencing credit access. This suggests that borrowers who asked for a longer credit duration are less likely to obtain microcredit. One possible explanation for this is that MFIs always consider the unexpected risks faced by rural borrowers during longer loan-terms in making their decision.

A binary logistic model is employed to identify determinants affecting the probability of being clients or non-clients of MFIs, estimating factors determining their participation as clients or non-clients. The binary logistic model reveals three significant determinants differentiating clients and non-clients of MFIs: the borrowers age (AGE), marital status (MAR) and educational attainment (EDU). In terms of the age of borrowers, the findings exhibit a significant and positive correlation between the age of households and the likelihood of being microcredit clients. The results indicate that older borrowers tend to become clients of MFIs. One possible explanation for this is that older households have better control over their household resources as they use microcredit in productive small business activities. More importantly, married householders have a greater probability of being clients of microcredit. This is because, in our surveyed study site, married borrowers are considered to have higher households incomes with two sources of income leading to their enhanced ability to repay a loan.

Finally, with regard to educational attainment, the result suggests that a higher level of educational attainment decreases the likelihood of being MFIs clients (at the 5% significance level). One potential explanation for this is that more highly educated borrowers tend gain better employment and prefer to choose formal financial sources. This would make borrowing from MFIs is unnecessary.

This study proposes a number of policy recommendations to government agencies/policy makers and microfinance practitioners/MFIs who have concerns about microfinance development in Indonesia. *The first recommendation* is related to the age of bor-

rowers. The findings revealed the significant impact of age on credit accessibility. Microcredit lenders tend to choose older borrowers rather than their vounger counterparts. One possible reason for this is that older borrowers are presumed to be mature and more experienced in managing their businesses (Anggraeni 2009). Hence, MFIs consider older borrowers to be more creditworthy. This implies that younger households suffered reduced microfinance access. To this extent, the policy makers should provide assistance to younger borrowers (e.g. basic skill training in business proposal, simple accounting report, and entrepreneurship management). In addition, MFIs might also consider focusing on the younger/start-up borrowers with supervision from field officers who have expertise and understand specific types of micro, small and medium enterprises (MSMEs). The purpose of this supervision would be to increase credit-worthiness of younger/start-up borrowers since MFIs regard them as low risk borrowers.

Second, this study reveals that MFIs preferred risktaking borrowers who willingly accept higher interest rates (Gray 2006). As a result, this will hinder non-risk taking borrowers who shy away from participating in microcredit. This leads to the issue of asymmetric information where microcredit lenders cannot identify who are good or bad borrowers. A possible approach in resolving this issue is for the government to continue efforts and build upon to preserve and back up the current microfinance insurance policy of Kredit Usaha Rakyat (KUR)/People's Business Credit (Finance Ministry 2016). Credit insurance has been implemented since the Indonesian Government launched Kredit Usaha Rakyat in 2008. This national program has given a mandate to several state-own banks and development regional banks (provincial owned-bank) to distribute small and soft loans to rural households (at 9% pa interest rates). The central government arranged for the Coordinating Ministry of Economic

Affairs to provide funds for guarantee fees and claim (up to 70% of the loan value) through *Asuransi Kredit Indonesia* (Askrindo) and *Jaminan Kredit Indonesia* (Jamkrindo) in case the rural borrowers face credit default<sup>8</sup>.

85

With regard to educational level, this study also found that microcredit lenders chose higher educated borrowers. Policy changes in response to this would be similar to the first recommendation; providing more practical assistance (e.g., information and technology training, and shared market information) to help less educated borrowers such that they would become more creditworthy to microcredit lenders.

Finally, in terms of loan duration, MFIs opted to select borrowers who applied for shorter loan terms. Many MFIs are concerned about the unexpected risks perceived to affect longer term loans by rural borrowers. To overcome this issue, microcredit insurance could be made available to MFIs to mitigate the risk of default by the borrowers. The credit insurance should be supported by central government fund (i.e. Askrindo and Jamkrindo). However, MFIs should also consider providing more diversified sources of funds, such as time-deposit products and saving accounts (e.g. Simpedes and Time Deposit of BRI). These products have enabled MFIs to become more resilient, sustainable and selfsufficient in making microcredit available to rural households (Seibel & Rachmadi 2009).

This study has several limitations. *The first limitation* is that the area of study only covered Bantul district in Yogyakarta, and thus limits our ability to generalize our results and findings. In addition, the length of field research was only two months (February–March 2016), which means that that there was limited time to gather in-depth data from the rural households.

<sup>&</sup>lt;sup>8</sup>Sources: www.askrindo.co.id and www.jamkrindo.co.id.

Economics and Finance in Indonesia Vol. 65 No. 1, June 2019

Future study should also take into account supplyside interviews with MFIs/microcredit lenders to investigate credit access in rural households. The purpose of investigating the supply-side sector is to get in-depth data for better impact assessment. The MNL model might also measure choices of financial sources accessed by rural households. Further, microfinance impacts studies should investigate different types of MFIs such as government aid program in several ministries (Agriculture and Fisheries), International Donors, Corporate Social Responsibility (CSR) fund and Islamic Microfinance Institutions. These non-bank MFIs have been serving microcredit for rural households in Indonesia with the same goal in poverty alleviation.

## References

- Abaru, B.M., Mugera, A.W., Norman, D.W. and Featherstone, A.M., 2006. The Uganda rural farmers scheme: women's accessibility to agricultural credit. *Agricultural Finance Review*, 66(2), pp.215-234. doi: https://doi.org/10.1108/00214660680001188.
- [2] Anggraeni, L., 2009. Factors influencing participation and credit constraints of a financial self-help group in a remote rural area: the case of ROSCA and ASCRA in Kemang village West Java. *Journal of Applied Sciences*, 9(11), pp.2067-2077.
- [3] Arsyad, L., 2005. An assessment of performance and sustainability of microfinance institutions: A case study of village credit institutions in Gianyar, Bali, Indonesia (PhD Thesis). Flinders University of Adelaide, Australia.
- [4] Ayamga, M., Sarpong, D.B. and Asuming-Brempong, S., 2006. Factors influencing the decision to participate in micro-credit programme: an illustration for Northern Ghana. *Ghana Journal of Development Studies*, *3*(2), pp.57-65. doi: http://dx.doi.org/10.4314/gjds.v3i2.35043.
- [5] Banking With The Poor, 2013. Indonesia Country Profile. Retrieved from http://www.bwtp.org/arcm/indonesia/ I\_Country\_Profile/Indonesia\_country\_profile.htm.
- [6] Chan, Y.H., 2005. Biostatistics 305. Multinomial logistic regression. *Singapore Medical Journal*, 46(6), pp.259-268.
- [7] Chaudhary, M.A. and Ishfaq, M., 2003. Credit worthiness of rural borrowers of Pakistan. *The Journal of Socio-Economics*, 32(6), pp.675-684. doi: https://doi.org/10.1016/j.socec.2003.10.005.

- [8] Churchill, C.F., 1999. Client-focused lending: the art of individual lending. Calmeadows.
- [9] Cochran, W.G., 2007. Sampling techniques. John Wiley & Sons.
- [10] Coleman, B.E., 2006. Microfinance in Northeast Thailand: Who benefits and how much?. *World Development*, 34(9), pp.1612-1638. doi: https://doi.org/10.1016/j.worlddev.2006.01.006.
- [11] Evans, T.G., Adams, A.M., Mohammed, R. and Norris, A.H., 1999. Demystifying nonparticipation in microcredit: a population-based analysis. *World Development*, *27*(2), pp.419-430. doi: https://doi.org/10.1016/S0305-750X(98)00134-X.
- [12] Finance Ministry., 2016. Interest rate lowered, government targets KUR of IDR100 trillion distributed. Jakarta. Retrieved from http://www.kemenkeu.go.id.
- [13] Getubig, M. and Gibbons, D., 2000. Financing a revolution: An overview of the microfinance challenge in Asia-Pacific. In B. Quinones & J. Remenyi (Eds.), *Microfinance and poverty alleviation: Case studies from Asia and the Pacific* (pp. 1–22), Routledge.
- [14] Gray, A., 2006. Credit accessibility of small-scale farmers and fisherfolk in the Philippines (Doctoral dissertation, Lincoln University). Retrieved from http://dspace.lincoln.ac.nz/ handle/10182/2255.
- [15] Greene, W.H., 2003. *Econometric analysis* (5th edition). Pearson Education. India.
- [16] Hermes, N. and Lensink, R., 2011. Microfinance: its impact, outreach, and sustainability. World Development, 39(6), pp.875-881. doi: https://doi.org/10.1016/j.worlddev.2009.10.021.
- [17] Ho, G., 2004. *Rural credit markets in Vietnam: Theory and practice* (Grand prize thesis). Macalester College.
- [18] Johnston Jr, D. and Morduch, J., 2008. The unbanked: evidence from Indonesia. *The World Bank Economic Review*, 22(3), pp.517-537. doi: https://doi.org/10.1093/wber/lhn016.
- [19] Karugia, J., Oluoch-Kosura, W., Nyikal, R., Odumbe, M. and Marenya, P., 2005. The role of rural factor markets in reducing poverty, risks and vulnerability in rural Kenya: The case of Kakamega and Vihiga Districts. Far East Journal of Theoretical Statistics; 2004.
- [20] Kashuliza, A.K. and Kydd, J.G., 1996. Determinants of bank credit access for smallholder farmers in Tanzania: a discriminant analysis application. *Savings and Development*, 20(3), pp.285-304.
- [21] Lebovics, M., Hermes, N. and Hudon, M., 2016. Are financial and social efficiency mutually exclusive? A case study of Vietnamese microfinance institutions. *Annals of Public and Cooperative Economics*, *87*(1), pp.55-77. doi: https://doi.org/10.1111/apce.12085.
- [22] Ledgerwood, J., 1998. Microfinance handbook: An insti-

*tutional and financial perspective.* The World Bank. doi: https://doi.org/10.1596/978-0-8213-4306-7.

- [23] Li, X., Gan, C. and Hu, B., 2011a. Accessibility to microcredit by Chinese rural households. *Journal of Asian Economics*, 22(3), pp.235-246. doi: https://doi.org/10.1016/j.asieco.2011.01.004.
- [24] Li, X., Gan, C. and Hu, B., 2011b. The welfare impact of microcredit on rural households in China. *The Journal of Socio-Economics*, 40(4), pp.404-411. doi: https://doi.org/10.1016/j.socec.2011.04.012.
- [25] Li, X., Gan, C. and Hu, B., 2011c. The impact of microcredit on women's empowerment: evidence from China. *Journal* of Chinese Economic and Business Studies, 9(3), pp.239-261. doi: https://doi.org/10.1080/14765284.2011.592352.
- [26] Mersland, R. and Strøm, R.Ø., 2008. Performance and trade-offs in Microfinance Organisations—Does ownership matter?. *Journal of International Development: The Journal* of the Development Studies Association, 20(5), pp.598-612. doi: https://doi.org/10.1002/jid.1432.
- [27] Miyata, S. and Sawada, Y., 2006. Credit accessibility, risk attitude, and social learning: Investment decisions of aquaculture in rural Indonesia. *Poster paper* prepared for presentation at the International Association of Agricultural Economists (IAAE) 2006 Annual Meeting, August 12-18, 2006, Queensland, Australia. Retrieved from https://ageconsearch.umn.edu/record/25669/.
- [28] Mohamed, K., (2003). Access to formal and quasi-formal credit by smallholder farmers and artisanal fishermen: A case of Zanzibar. *Research Report No. 03.6*. Mkuki na Nyota Publishers. Retrieved from http://www.repoa.or.tz/ documents/03.6\_-\_Mohamed\_.pdf.
- [29] Morduch, J., 1998. Does microfinance really help the poor?: New evidence from flagship programs in Bangladesh. Research Program in Development Studies, Woodrow School of Public and International Affairs.
- [30] Morduch, J., 1999. The microfinance promise. *Journal of Economic Literature*, *37*(4), pp.1569-1614. doi: 10.1257/jel.37.4.1569.
- [31] Nagarajan, G. and Meyer, R.L., 1996. Collateral Substitutes: Effect on Loan Access and Size in the Philippine Informal Credit Markets. *Selected paper* presented at the meetings of the 1996 American Agricultural Economics Association, Economics and Sociology Occasional Paper No. 2324, Department of Agricultural Economics and Rural Sociology, The Ohio State University, Columbus, July 1996.
- [32] Nouman, M., Siddiqi, M., Asim, S. and Hussain, Z., 2013. Impact of socio-economic characteristics of farmers on access to agricultural credit. *Sarhad Journal of Agriculture*, 29(3), pp.469-476.
- [33] Okten, C. and Osili, U.O., 2004. Social networks and credit access in Indonesia. *World Development*, 32(7), pp.1225-1246. doi: https://doi.org/10.1016/j.worlddev.2004.01.012.

- [34] Okurut, F.N., 2006. Access to credit by the poor in South Africa: evidence from household survey data 1995 and 2000. Stellenbosch Economic Working Papers 13/06. Stellenbosch: University of Stellenbosch. Retrieved from https://www.ekon.sun.ac.za/wpapers/2006/ wp132006/wp-13-2006.pdf.
- [35] Panjaitan-Drioadisuryo, R.D. and Cloud, K., 1999. Gender, self-employment and microcredit programs an Indonesian case study. *The Quarterly Review* of Economics and Finance, 39(5), pp.769-779. doi: https://doi.org/10.1016/S1062-9769(99)00028-9.
- [36] Prawiranata, I.R., 2013. Sustainable microfinance in Indonesia: A sociocultural approach (Doctoral dissertation, Victoria University). Retrieved from http://vuir.vu.edu.au/ 22309/.
- [37] Quinones, B. and Remenyi, J., 2000. Microfinance and poverty alleviation: Case studies from Asia and the Pacific. Routledge.
- [38] Raleting, P.M. and Obi, A., 2015. An analysis of institutional factors influencing vegetable production amongst smallscale farmers in six vegetable projects of the Nkonkobe Local Municipality. *Journal of Agricultural Science*, 7(6), pp.184-196. doi: 10.5539/jas.v7n6p184.
- [39] Ravi, S., 2004. Access to credit and role of co-operation. Retrieved from http://neumann.hec.ca/neudc2004/fp/ravi\_ shamika\_sept\_27.pdf.
- [40] Revindo, M.D. and Gan, C., 2017a. Rural microfinance banking viability and outreach: A case of bank Rakyat Indonesia. In C. Gan & G. V. Nartea, *Microfinance in Asia*, pp.337-356. World Scientific Pub.
- [41] Revindo, M.D. and Gan, C., 2017b. Microfinance institutions in Malaysia. In C. Gan & G. V. Nartea, *Microfinance in Asia*, pp.47-91. World Scientific Pub.
- [42] Revindo, M.D. and Gan, C., 2017c. Credit market and microcredit services in rural China. In C. Gan & G. V. Nartea, *Microfinance in Asia*, pp.93-146. World Scientific Pub.
- [43] Robinson, M., 2002. The microfinance revolution: Lessons from Indonesia. The World Bank. doi: https://doi.org/10.1596/0-8213-4953-8.
- [44] Rosengard, J.K. and Prasetyantoko, A., 2011. If the banks are doing so well, why can't I get a loan? Regulatory constraints to financial inclusion in Indonesia. *Asian Economic Policy Review*, 6(2), pp.273-296. doi: https://doi.org/10.1111/j.1748-3131.2011.01205.x.
- [45] Rosengard, J.K., Patten, R.H., Johnston Jr, D.E. and Koesoemo, W., 2007. The promise and the peril of microfinance institutions in Indonesia. *Bulletin of Indonesian Economical Studies*, 43(1), pp.87-112. doi: https://doi.org/10.1080/00074910701286404.
- [46] Saad, N.M. and Duasa, J., 2011. An economic impact assessment of a microcredit program in Malaysia: The case of Amanah Ikhtiar Malaysia (AIM). *International Journal of*

Business & Society, 12(1), pp.1-14.

88

- [47] Sarap, K., 1990. Factors affecting small farmers' access to institutional credit in rural Orissa, India. Development and Change, 21(2), pp.281-307. doi: https://doi.org/10.1111/j.1467-7660.1990.tb00378.x.
- [48] Scott Long, J., 1997. Regression models for categorical and limited dependent variables. Advanced Quantitative Techniques in the Social Sciences 7. Thousand Oaks: Sage Publications.
- [49] Sebopetji, T.O. and Belete, A., 2009. An application of probit analysis to factors affecting small-scale farmers decision to take credit: A case study of the Greater Letaba Local Municipality in South Africa. *African Journal of Agricultural Research, 4*(8), pp.718-723.
- [50] Seibel, H.D. and Parhusip, U., 1998. Attaining outreach with sustainability: A case study of a private micro-finance institution in Indonesia. *IDS Bulletin*, 29(4), pp.81-90. doi: https://doi.org/10.1111/j.1759-5436.1998.mp29004009.x.
- [51] Seibel, H.D. and Rachmadi, A., 2009. Growth and resilience of savings-based microfinance institutions: The case of the microbanking units of Bank Rakyat Indonesia. Retrieved from: http: //www.arabic.microfinancegateway.org/sites/default/files/ mfg-en-case-study-growth-and-resilience-of-savingsbased-microfinance-institutions-the-case-of-themicrobanking-units-of-bank-rakyat-indonesia-2009.pdf.
- [52] Tambunan, T., 2015. Financial inclusion, financial education, and financial regulation: A story from Indonesia. ADBI Working Paper Series No. 535. Asian Development Bank Institute, Tokyo, Japan. Retrieved from https://www.adb.org/ sites/default/files/publication/161176/adbi-wp535.pdf.
- [53] Tang, S., Guang, Z. and Jin, S., 2010. Formal and informal credit markets and rural credit demand in China. Selected paper prepared for presentation at the Agricultural & Applied Economics Association 2010 AAEA,CAES, & WAEA Joint Annual Meeting, Denver, Colorado, July 25-27, 2010. Retrieved from https: //ageconsearch.umn.edu/bitstream/61339/2/Formal% 20and%20Informal%20Credit%20Markets%20and% 20Rural%20Credit%20Demand%20in%20China.pdf.
- [54] Train, K., 1986. Qualitative choice analysis: Theory, econometrics, and an application to automobile demand. MIT press.
- [55] Tsukada, K., Higashikata, T. and Takahashi, K., 2010. Microfinance penetration and its influence on credit choice in Indonesia: evidence from a household panel survey. *The Developing Economies*, 48(1), pp.102-127. doi: https://doi.org/10.1111/j.1746-1049.2010.00100.x.
- [56] Umoh, G.S., 2006. Empirical investigation of access to micro-credit in an emerging economy: Evidence from Nigeria. *Journal of African Business*, 7(1-2), pp.89-117. doi: https://doi.org/10.1300/J156v07n01\_05.

- [57] Vaessen, J., 2001. Accessibility of rural credit in Northern Nicaragua: The importance of networks of information and recommendation. *Savings and Development*, 25(1), pp.5-32.
- [58] Wijesiri, M., Viganò, L. and Meoli, M., 2015. Efficiency of microfinance institutions in Sri Lanka: a two-stage double bootstrap DEA approach. *Economic Modelling*, 47, pp.74-83. doi: https://doi.org/10.1016/j.econmod.2015.02.016.
- [59] World Bank., 2009. Improving access to financial services in Indonesia: Executive summary. Retrieved from http://documents.worldbank.org/curated/ en/448491468257952510/Executive-summary.
- [60] Wydick, B., Hayes, H.K. and Kempf, S.H., 2011. Social networks, neighborhood effects, and credit access: Evidence from rural Guatemala. *World Development*, 39(6), pp.974-982. doi: https://doi.org/10.1016/j.worlddev.2009.10.015.
- [61] Zeller, M., 1994. Determinants of credit rationing: A study of informal lenders and formal credit groups in Madagascar. *World Development*, 22(12), pp.1895-1907. doi: https://doi.org/10.1016/0305-750X(94)90181-3.